Author's response to reviews

Title: Hepatitis B and C: Prevalence and social factors associated with seropositivity among children in Karachi, Pakistan

Authors:

Wasim DR Jafri (wasim.jafri@aku.edu)
Nadim DR Jafri (nadimj@hotmail.com)
Javed DR Yakoob (yakoobjaved@hotmail.com)
Muhammad MR Islam (muhammad.islam@aku.edu)
Syed DR Tirmizi (syed.tirmizi@aku.edu)
Tazeen DR Jafar (tazeen.jafar@aku.edu)
Saeed DR Akhtar (saeed.akhtar@aku.edu)
Saeed DR Hamid (saeed.hamid@aku.edu)
Hasnain DR Alishah (hasnain.alishah@aku.edu)
Sheikh Qamaruddin DR Nizami (qamaruddin.nizami@aku.edu)

Version: 2 Date: 28 March 2006

Author's response to reviews:

27 March, 2005

Dear Sir

This paper should be published in BMC Infectious Diseases as it has highlighted the prevalence and social factors associated with Hepatitis B and C in a juvenile population of third world country. Hepatitis B and C virus infections are common in the developing countries with an early age of chronic liver disease and hepatocellular carcinoma.

Thanking you.

Yours sincerely,

Dr Wasim Jafri FACP, FACG, FRCP
Chair, Faculty of Medicine
Chief of Section of Gastroenterology,
Aga Khan University Hospital
Karachi-74800
Pakistan